Author's response to reviews

Title: Is the presence of Modic changes associated with the outcomes of different treatments? A systematic critical review

Authors:

Rikke K Jensen (rikke.kruger.jensen@slb.regionsyddanmark.dk)
Charlotte Leboeuf-Yde (clyde@health.sdu.dk)

Version: 3 Date: 28 July 2011

Author's response to reviews: see over
Dear Editor,

Thank you for accepting our manuscript “Is the presence of Modic changes associated with the outcomes of different treatments? A systematic critical review”. Our changes have been highlighted in the manuscript.

On the behalf of the research team,

Rikke Krüger Jensen
Research Department
Spine Centre of Southern Denmark
Oestre Hougvej 55
DK - 5500 Middelfart
Denmark